Author's response to reviews

Title: Validation of human papillomavirus genotyping by signature DNA sequence analysis

Authors:

Sin Hang Lee (sinhang.lee@milfordhospital.org)
Veronica S Vigliotti (veronica.vigliotti@milfordhospital.org)
Jessica S Vigliotti (09_jvigliotti@stagweb.fairfield.edu)
Suri Pappu (suri.pappu@milfordhospital.org)

Version: 5 Date: 1 May 2009

Author's response to reviews:

May 1, 2009
To: The Editor
BMC Clinical Pathology

Re: Validation of human papillomavirus genotyping by signature DNA sequence analysis
-minor revision

Dear Editor:

Thank you and the reviewer for considering acceptance of the above referenced manuscript for publication in BMC Clinical Pathology. On behalf of the co-authors, I have made the following changes:

1) Per advice of the reviewer, the sequence of the 23 bases of the GP5+ priming site in Figure 1 has been underlined.

2) One word “underlined” is added to the legend for Figure 1 in the manuscript, reflecting this change in #1).

3) The online link to Reference #28, which is recently published, is added.

Thank you for your further consideration of this manuscript and please contact the undersigned if additional revisions are needed.

Sincerely,

Sin Hang Lee, MD